Author's response to reviews

Title: Measures of upper limb function for people with neck pain: A systematic review of measurement and practical properties (Protocol)

Authors:

Ahmad SE Alreni (ahmadalreni@hotmail.co.uk)
Deborah Harrop (d.harrop@shu.ac.uk)
Anil Gumber (a.gumber@shu.ac.uk)
Sionnadh McLean (s.mclean@shu.ac.uk)

Version: 3 Date: 19 March 2015

Author's response to reviews: see over
Measures of upper limb function for people with neck pain: A systematic review of measurement and practical properties (Protocol)

Point-by-point response to the Editor concerns

Concern – 1
“Cochran is missing an “e”

Response
The missing letter “e” has now been added to Cochrane and indicated in red text.

Concern – 2
The first c in “Cochrane” should be capitalised; likely the same with “collaboration”.

Response
The “c” has now been capitalised and indicated in red text.